
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Tue Oct 17 16:56:16 GMT 2006
Year: 06  Doy: 290
Observer: koon
WEATHER COMMENT: Tue Oct 17 16:56:17 GMT 2006
Foggy, overcast, wind=20 mph from the South, temp=42F.
___end___
COMMENT: Tue Oct 17 16:56:50 GMT 2006
No obs until fog and clouds clear enough.
___end___
WEATHER COMMENT: Tue Oct 17 16:57:56 GMT 2006
Thunderstorms last night, apparent power outage here since yesterday obs.
___end___
**GONG PROBLEM**: Tue Oct 17 17:16:22 GMT 2006
Cycled power to Media Converter in shelter to reconnect between us.
___end___
Tue Oct 17 19:37:49 GMT 2006    CHIP     Start Patrol
Tue Oct 17 19:38:04 GMT 2006    PICS     Start Patrol
Tue Oct 17 19:38:25 GMT 2006    MKIV    Start Patrol
WEATHER COMMENT: Tue Oct 17 19:38:26 GMT 2006
Clouds thinned, starting some instruments.
___end___
Tue Oct 17 19:44:40 GMT 2006    MKIV    End Patrol
PSPT COMMENT: Tue Oct 17 19:49:55 GMT 2006
obs.
___end___
Tue Oct 17 22:16:56 GMT 2006    PICS     End Patrol
PICS COMMENT: Tue Oct 17 22:17:00 GMT 2006
Pausing to put in a larger occulting disk at Joan’s request.
___end___
Tue Oct 17 22:21:39 GMT 2006    PICS     Start Patrol
PICS COMMENT: Tue Oct 17 22:21:38 GMT 2006
Changed occulter from 1981" to 1991".
___end___
NICE PICSDISC IMAGE: 2038
NICE PICSLIMB IMAGE: 2130
NICE CHIP     IMAGE: 2146
WEATHER COMMENT: Wed Oct 18 01:12:25 GMT 2006
Lower clouds are passing over.
___end___
Wed Oct 18 01:26:30 GMT 2006    CHIP     End Patrol
Wed Oct 18 01:26:51 GMT 2006    PICS     End Patrol
WEATHER COMMENT: Wed Oct 18 01:27:19 GMT 2006
Pausing instruments due to clouds.
___end___
**PSPT PROBLEM**: Wed Oct 18 01:48:14 GMT 2006



Stalled while shutting down, pointed dome with "dome goto" command.
___end___
Wed Oct 18 01:54:42 GMT 2006 
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